Author's response to reviews

Title: Design of a Randomized Controlled Trial for Multiple Cancer Risk Behaviors among Spanish-speaking Mexican-Origin Smokers

Authors:

Yessenia Castro (ycastro@austin.utexas.edu)
Karen Basen-Engquist (kbasenen@mdanderson.org)
Maria E Fernandez (Maria.E.Fernandez@uth.tmc.edu)
Larkin L Strong (LLstrong@mdanderson.org)
Elizabeth G Eakin (e.eakin@uq.edu.au)
Ken Resnicow (kresnic@umich.edu)
Yisheng Li (ysli@mdanderson.org)
David W Wetter (Dwetter@mdanderson.org)

Version: 2 Date: 22 February 2013

Author's response to reviews:

Dear Dr. Silvestre,

Thank you for the input following a preliminary review of our manuscript. We apologize for the omission of important information relevant to the study protocol. Please see our responses below:

-- The study is supported by external funding, and the grant did undergo a peer review process. Funding is from the National Cancer Institute, National Institutes of Health, U54CA153505. The amount awarded was $XX.

-- The protocol also received approval from the Institutional Review Board at the UT MD Anderson Cancer Center

-- No other manuscripts have been published or submitted that are based on this protocol

-- We have added the Trial Registration Number at the end of the Abstract and have included in the Methods section how written informed consent was obtained.

Please do not hesitate to contact me with additional questions.

Sincerely,

Larkin Strong